Author’s response to reviews

Title: Use of ivacaftor in late diagnosed Cystic Fibrosis monozygotic twins heterozygous for F508del and R117H-T – a case report

Authors:

Matthias Welsner (matthias.welsner@rlk.uk-essen.de)
Svenja Straßburg (svenja.steiner@rlk.uk-essen.de)
Christian Taube (christian.taube@rlk.uk-essen.de)
Sivagurunathan Sutharsan (sivagurunathan.sutharsan@rlk.uk-essen.de)

Version: 2 Date: 10 Mar 2019

Author’s response to reviews:

Dear Editor, dear Reviewers

We carried out an extensive revision of our original manuscript on the suggestions of previous reviews. In addition to changes in the text, the abstract was also reworked. The comments that led to the revision of the manuscript were answered in the cover letter.

Yours sincerely

Matthias Welsner